BEST CLINICAL PRACTICES

CHAPTER 13

from the

INTERNATIONAL POSITION PAPER

0 N

Women's Health and Menopause: A Comprehensive Approach

National Heart, Lung, and Blood Institute Office of Research on Women's Health National Institutes of Health and Giovanni Lorenzini Medical Science Foundation

BEST CLINICAL PRACTICES CHAPTER 13 from the INTERNATIONAL POSITION PAPER on

Women's Health and Menopause: A Comprehensive Approach

NATIONAL HEART, LUNG, AND BLOOD INSTITUTE

NIH OFFICE OF RESEARCH ON Women's Health

GIOVANNI LORENZINI MEDICAL Science Foundation

For Administrative Use Only March 2002

PREFACE

Women's health and menopause is a rapidly expanding field of medical practice and scientific investigation. It is a field of great social importance and impact, nationally and globally in developed as well as developing countries.

Menopause is a normal event in a woman's life. Some women view it as a positive and liberating experience. Others think of it as a negative event. Today, most women live long enough to become postmenopausal. In the developed world, the percentage of women over 50 years of age has tripled in the last 100 years. During this period, women's life expectancy in the United States has increased from 50 to 81.7 years, meaning that more than one-third of life will be lived in postmenopause.

This chapter on Best Clinical Practices summarizes the evidence-based recommendations from the International Position Paper on Women's Health and Menopause: A Comprehensive Approach, to be published by the National Heart, Lung, and Blood Institute (NHLBI), National Institutes of Health (NIH) in the spring of 2002. The International Position Paper is based on extensive international review and evaluation of the scientific evidence for current clinical practices as presented in the published literature. The purpose of this international and multidisciplinary monograph is to enhance the composite health of menopausal and postmenopausal women on a global basis, and to make recommendations regarding needs for future research.

The International Position Paper represents the culmination of 7 years of cooperation between the NHLBI and the Giovanni Lorenzini Medical Science Foundation (Milan, Italy and Houston, TX, USA) in a public/private partnership in the development and cosponsorship of four international conferences on Women's Health and Menopause since the mid-1990s. The first three conferences were held in Italy, and the most recent one in May 2001 in Washington, D.C. The last two conferences were also cosponsored by the NIH Office of Research on Women's Health. These conferences have addressed not only cardiovascular disease (CVD), but other health problems such as cancer, osteoporosis, and Alzheimer's disease, as well as the use and impact of hormone replacement therapy.

On behalf of the National Heart, Lung, and Blood Institute, I would like to thank Nanette K. Wenger, M.D., Chair of the Executive Committee of the International Position Paper; Rodolfo Paoletti, M.D., Cochair; Vivian W. Pinn, M.D., Cochair; the Panel Members; and the referees and reviewers for their valuable scientific contributions and dedication during this 4-year effort to critically review and evaluate extensive international databases on women's health and menopause. I hope that the readers will come away with a sense of appreciation for what biomedical research has accomplished in this important field and anticipation of the many opportunities and needs for future research.

C. Indon

Claude Lenfant, M.D.

Director, National Heart, Lung, and Blood Institute *and Cochair*, Executive Committee of the International Position Paper on Women's Health and Menopause: A Comprehensive Approach

INTERNATIONAL POSITION PAPER ON WOMEN'S HEALTH AND MENOPAUSE: A COMPREHENSIVE APPROACH

EXECUTIVE COMMITTEE AND PANEL

Nanette K. Wenger, M.D., USA served as the Chair of the Panel and Editor of the monograph. The Cochairs and Coeditors were Rodolfo Paoletti, M.D., Italy; Claude J.M. Lenfant, M.D., USA; and Vivian W. Pinn, M.D., USA

THE PANEL MEMBERS INCLUDED:

Elizabeth Barrett-Connor, M.D., USA Martin H. Birkhäuser, M.D., Switzerland Louise A. Brinton, Ph.D., USA Aila Collins, Ph.D., Sweden Peter Collins, M.D., UK Pier Giorgio Crosignani, M.D., Italy Lorraine Dennerstein, A.O., M.B.B.S., Ph.D., D.P.M., FRANZCP, Australia Bruce Ettinger, M.D., USA Jan-Åke Gustafsson, M.D., Ph.D., Sweden Victor W. Henderson, M.D., M.S., USA Susan Hendrix, D.O., USA Carlo LaVecchia, M.D., Italy Robert Lindsay, Ph.D., USA Adriana Maggi, Ph.D., Italy Joan A. McGowan, Ph.D., USA Anne McTiernan, M.D., Ph.D., USA Maryann Redford, D.D.S., M.P.H., USA Susan M. Resnick, Ph.D., USA Jacques E. Rossouw, M.D., USA Sherry Sherman, Ph.D., USA

NONPANEL COAUTHORS

Additional authors who were invited to contribute to individual chapters:

Janet R. Guthrie, Ph.D., Australia	Stefan Nilsson, Dr. Med. Sci., Sweden
Barbara E.K. Klein, M.D., M.P.H, USA	Nanette Santoro, M.D., USA

SPECIAL ACKNOWLEDGMENTS

The Panel and the clinical community owe a debt of gratitude to the referees and reviewers for their expertise, for their thoughtful comments and additions to the initial version of the chapters, and for their dedication to the enhancement of menopausal health.

THE EXPERTS WHO REVIEWED THE PRELIMINARY INDIVIDUAL CHAPTERS INCLUDED:

Beth Alder, Ph.D., UK Gloria A. Bachmann, M.D., USA Linda Cardozo, M.D., UK Sir Richard Doll, UK Timothy M.M. Farley, Ph.D., Switzerland Marcha Flint, Ph.D., USA Silvia Franceschi, M.D., France Ulysses J. Gaspard, M.D., Ph.D., Belgium Anna Glasier, M.D., Ph.D., UK Elina Hemminki, M.D., Ph.D., Finland Myra S. Hunter, Ph.D., UK Won-Whe Kim, M.D., Korea Britt-Marie Landgren, M.D., Ph.D., Sweden Lenore Manderson, Ph.D., Australia Olav Meirik, M.D., Ph.D., Chile Daniel R. Mishell Jr., M.D., USA

Bert O'Malley, M.D., USA Bent Ottesen, M.D., Denmark David Purdie, M.D., UK Lawrence G. Raisz, M.D., USA Henri Rochefort, M.D., Ph.D., France Giuseppe Rosano, M.D., Ph.D., Italy Raymond Rosen, Ph.D., USA Karin Schenk-Gustafsson, M.D., Ph.D., Sweden Peter Schmidt, M.D., USA Ethel Siris, M.D., USA Sven O, Skouby, M.D., Denmark Leon Speroff, M.D., USA John C. Stevenson, M.B.B.S., UK David W. Sturdee, M.D., UK Cornelia Van Duijn, M.D., The Netherlands Nancy Woods, Ph.D., R.N., USA

THE EXPERTS WHO REVIEWED THE PRELIMINARY COMPOSITE DOCUMENT WERE:

Takeshi Aso, M.D., Ph.D., Japan Giuseppe Benagiano, M.D., Italy Henry Burger, M.D., Ph.D., Australia Desmond Julian, M.D., UK Peter Kenemans, M.D., Ph.D., The Netherlands Luella Klein, M.D., USA Rogerio Lobo, M.D., USA Michael Mendelsohn, M.D., USA Henri Rozenbaum, M.D., France Goran Samsioe, M.D., Sweden Isaac Schiff, M.D., USA Piero Sismondi, M.D., Ph.D., Italy Wulf H. Utian, M.D., USA Nelson Watts, M.D., USA

Special acknowledgment is warranted to Suzanne S. Hurd, M.D., USA for her initial chairing of the project and for her scientific organizational skills; and to Ruth Johnsson Hegyeli, M.D., USA for serving as the liaison between the Panel and the National Institutes of Health, providing invaluable expert counsel. Also acknowledged with gratitude are the contributions of Silvia Belcredito, junior editor of the monograph, Andrea Cignarella, and Adriana Maggi of the Department of Pharmacological Sciences of the University of Milan, Italy; Cristina Bolsi, Italy; Anna Cozzi, Italy; Emanuela Folco, Italy; Ann S. Jackson, USA; Anna Miniotti, Italy, of the Giovanni Lorenzini Medical Foundation (Milan, Italy and Houston, TX, USA); and Nancy Morris of the National, Heart, Lung and Blood Institute (NHLBI) for their coordinating and editorial assistance.

CHAPTER 13 FROM THE INTERNATIONAL POSITION PAPER ON WOMEN'S HEALTH AND MENOPAUSE: A COMPREHENSIVE APPROACH

Elizabeth Barrett-Connor, M.D.*; Susan Hendrix, D.O.**; Bruce Ettinger, M.D.***

Nanette K. Wenger, M.D. Chair; Rodolfo Paoletti, M.D., Cochair; Claude J.M. Lenfant, M.D., Cochair; Vivian W. Pinn, M.D., Cochair

Panel: Martin H. Birkhäuser, M.D.; Louise A. Brinton, Ph.D.; Aila Collins, Ph.D.; Peter Collins, M.D.; Pier Giorgio
Crosignani, M.D.; Lorraine Dennerstein, A.O., M.B.B.S., Ph.D., D.P.M., FRANZCP; Jan-Åke Gustafsson, M.D., Ph.D.;
Victor W. Henderson, M.D., M.S.; Carlo La Vecchia, M.D.; Robert Lindsay, Ph.D.; Adriana Maggi, Ph.D.; Joan A.
McGowan, Ph.D.; Anne McTiernan, M.D., Ph.D.; Maryann Redford, D.D.S., M.P.H.; Susan M. Resnick, Ph.D.; Jacques E.
Rossouw, M.D.; Sherry Sherman, Ph.D.

1. INTRODUCTION

The proportion of women living past the age of menopause has tripled during the past century, and is expected to increase steadily in the foreseeable future. If adulthood is defined as beginning at age 21, the average age at menopause as age 51, and the average life expectancy as age 81, women in the United States, in Europe, and in much of the developed world will live one-half their adult lives in the years after menopause, a time of relative estrogen deficiency compared to their reproductive years.

In recent years, the aging of the female population, together with the availability of "replacement" hormones, led to numerous studies of the menopause. Most of these studies were of middle-class white women living in the United States and Western Europe, with results that may not be relevant to other women. In addition, many studies were clinical or epidemiological observations of associations—less satisfactory for evidence-based medicine than randomized, placebo-controlled, double-blind clinical trials. Results from recent clinical trials studying the benefits of hormone replacement therapy (HRT) have differed from observational studies. Results from additional large clinical trials, expected in the next 5 years, may further change thinking about the optimal management for the menopausal woman.

The menopause offers the health care provider an opportunity to assess each woman's health, her concerns, and the need for health promotion and disease prevention measures. Today's health care provider has to consider a bewildering array of

^{*}From the Division of Epidemiology, Department of Family and Preventive Medicine, UCSD School of Medicine, La Jolla, California, United States.

^{**} From the Division of Gynecology, Department of Obstetrics and Gynecology, Wayne State University School of Medicine, Hutzel Hospital, Detroit, Michigan, United States.

^{***}From the Division of Research, Kaiser Permanente Medical Care Program, Oakland, California, United States.

changing "facts" and sees increasingly informed patients with strong personal convictions about the menopause and their need for medication. The provider must be prepared to discuss a variety of menopause or age-related topics, and decide what to recommend for a specific woman—often in less time than ever before.

The menopause offers the health care provider an opportunity to assess each woman's health, her concerns, and the need for health promotion and disease prevention measures. Recommendations should be specific to each woman and her background. There are country-specific and cultural variations in menopause symptoms, the frequency of different postmenopausal diseases, clinical practice, health care resources, and afford-

able interventions. Country- and culture-specific practice will therefore vary, and appropriately so.

Recent research findings include:

- Increasing recognition of the need to address health promotion beyond the perimenopausal years, and in women with and without menopausal symptoms.
- The risks and benefits of lifestyle, pharmacological, and surgical interventions may change as women age.
- The tailoring of menopausal treatment to the individual woman should be based on her individual clinical profile and concerns.
- For the treatment of the climacteric syndrome, HRT remains the most effective pharmacologic intervention.
- The long-term benefits and risks of HRT continue to be assessed.

- HRT for long-term health promotion, as for osteoporosis, usually requires its continued use.
- New nonhormone therapies offer more options for prevention of common menopausal conditions, to be tested fully in clinical trials.
- Preventive drug therapy can start many years after menopause, particularly with respect to osteoporosis. This, however, may not be optimal.
- The risk for many disease outcomes can be reduced even in old age.

Research results are teaching us to be cautious before assuming that current practice is best. For example, although HRT remains the gold standard for the treatment of vasomotor and urogenital symptoms, as well as for the prevention of bone loss, recent clinical trial results have failed to show benefit for other menopausal conditions, such as incontinence.

Past perceptions about appropriate indications for the use of HRT were based almost entirely on clinical experience and observational data. These perceptions are being questioned as new knowledge emerges from clinical trials. Some examples follow:

The long-term benefits and risks of HRT continue to be assessed.

- *Perception:* HRT protects against coronary heart disease.
- *Evidence:* The prospective randomized clinical trials reported so far have not shown benefit for reducing coronary events in secondary prevention.

Terception.	retards the progression of Alzheimer's disease.
Evidence:	Clinical trials have not shown that estrogen replacement therapy retards progression of early Alzheimer's disease. Clinical trials of the effects of estrogen replacement therapy on memory loss are still ongoing.
Perception:	Estrogen improves symptoms of depression.
Evidence:	Clinical trial data have shown that estrogen replacement therapy improves mood and well-being only in women with vasomotor symptoms and sleep disturbance. No convinc-

Darcantion: Estrogen provents memory loss and

ing clinical trial data indicate that estrogen therapy in postmenopausal women is an effective treatment for major depression.

- *Perception:* Estrogen improves urinary incontinence.
- *Evidence:* Clinical trials have shown no benefit.
- *Perception:* HRT, unlike oral contraceptives, does not increase the risk of venous thromboembolism.
- *Evidence:* Clinical trials confirm a threefold increased risk of venous thromboembolism with oral estrogen replacement therapy.

- **Perception:** HRT during the first 5–10 years after the menopause is sufficient to prevent osteoporosis in later life.
- *Evidence:* Bone loss resumes after stopping HRT, leaving women vulnerable to osteoporosis in later life. Whether estrogen is started early or late, it must be continued into old age to maintain skeletal health.

Although many clinically relevant questions remain unanswered, women seeking advice about the menopause now have more information about and more options for healthy postmenopausal years

New nonhormone therapies offer more options for prevention of common menopausal conditions, to be tested fully in clinical trials.

than ever before. New trial results and new medications may further change recommendations for the assessment and management of the postmenopausal woman.

We make the following recommendations concerning HRT based on results from clinical trials. Where evidence from clinical trials is not available, we make recommendations based on observational studies. These recommendations are intended as guidelines, not mandates. All interventions should be individualized—tailored to the specific needs and concerns of each woman and designed to provide an optimal quality of life.

2. THE MENOPAUSE TRANSITION

2.1 Assessment

Some women sail through the menopause transition with no complaints, others are miserable, and the majority have symptoms that are somewhat bothersome. There is still controversy as to which symptoms are related to menopause and which are associated with or exacerbated by other factors.

Clearly not all symptoms that occur during the menopause transition are due to hormonal changes. Only vasomotor, sleep, and some vulvovaginal symptoms have shown more favorable relief after HRT than placebo, and can therefore be convincingly attributed to changing hormone levels and menopause. At least 25 percent of women in clinical trials report significant improvement in their vasomotor symptoms when taking placebo. Therefore, convincing evidence of treatment benefit requires a placebo-controlled clinical trial.

Women may visit the physician because they have symptoms they suspect are related to menopause or because they want information about the menopause. In either case, assessment for symptoms provides an opportunity to discuss issues that might otherwise remain unaddressed.

Symptoms directly or indirectly related to the menopause transition include:

- Vasomotor symptoms (hot flushes and night sweats)
- Sleep-related symptoms
- Mood changes
- Sexual dysfunction

- Problems with concentration and memory
- Urogenital symptoms

Symptoms can be queried by using a questionnaire or symptom checklist during the interview. Some find that asking about sexual satisfaction is facilitated by the use of a checklist. When a checklist is used, it is important to go beyond the list to assess the severity and duration of reported symptoms and the degree to which they interfere with the woman's life. For example, hot flushes that are not bothersome do not require treatment.

When asking about symptoms, the clinician should be sensitive to each woman's:

- Beliefs and attitudes about menopause, including her medical vs. nonmedical treatment preferences, anxieties, and coping style
- Sociocultural and ethnic background that may affect her concerns and choices
- Work situation, job satisfaction, and stress
- Other life stressors, particularly with personal relationships
- Social supports
- Overall quality of life
- Current use of nonprescription herbal, nutriceutical (a nutritional supplement designed for a specific clinical purpose), or phytoestrogen remedies.

It is important to have a dialogue with the patient. Failure to listen and discuss may explain why many women prescribed HRT do not fill the prescription.

2.2 Symptom Prevention and Treatment. Vasomotor Symptoms: Hot Flushes and Night Sweats

2.2.1 Lifestyle

- Wear layered clothing that can be removed or added as necessary.
- There is conflicting evidence as to whether exercise improves menopause vasomotor symptoms.

2.2.2 Diet

- Avoid hot spicy foods and beverages, and reduce caffeine.
- Avoid alcohol beverages (excess can cause flushing).

2.2.3 Pharmacotherapy

- In a systematic review of more than 40 randomized controlled clinical trials, oral and transdermal estrogen each reduced the severity of vasomotor symptoms, and estrogen was effective in doses lower than the usual 0.625 mg of equine estrogens or equivalent.
- Transdermal estradiol and intranasal 17-beta estradiol spray are as effective as oral estrogen in reducing hot flushes.
- Oral tibolone is as effective as other forms of HRT such as estradiol valerate or conjugated estrogens in reducing hot flushes.
- Selective estrogen receptor modulators can increase hot flushes. In clinical trials, approximately 20 percent of women at least 2 years after menopause less than 60 years of age and 10 percent of older women developed hot flushes on raloxifene. Vasomotor symptoms were mild and rarely led to discontinuation of therapy.
- The selective serotonin reuptake inhibitors (SSRIs) venlafaxine and paroxetine have been shown to substantially reduce hot flushes in clinical trials.

- Progestogens in high daily doses (medroxyprogesterone acetate 20 mg per day or megestrol acetate 40 mg per day) also reduced vasomotor symptoms.
- Veralipride (100 mg per day) reduces hot flushes in patients treated with GnRH agonists.
- Propranolol is no more effective than placebo for the reduction of hot flushes, whereas evidence for clonidine's benefit is inconsistent.

2.2.4 Complementary and Alternative Therapies

• Phytoestrogens have not been shown in most clinical trials to decrease vasomotor symptoms significantly better than placebo. Different results may relate to differences in women (not all of them absorb phytoestrogens equally well) or differences in the products tested. The best single dietary source of phytoestrogens

is soy. The U.S. Food and Drug Administration has approved a statement that soy protein at a dose of 25 gm/day may reduce the risk of CVD, based on a modest reduction in total cholesterol level.

It is important to have a dialogue with the patient. Failure to listen and discuss may explain why many women prescribed HRT do not fill the prescription.

- Dong quai has been shown in a clinical trial not to be more effective than placebo for the treatment of hot flushes.
- Evening primrose oil (gamma-linolenic acid) is not more effective than placebo for the reduction of hot flushes.

2.3 Symptom Prevention and Treatment. Urogenital Symptoms

• At least nine randomized controlled trials have shown that estrogen improves urogenital symptoms; this is true for oral and transdermal estrogen and for a silicone estradiol releasing vaginal ring. Vaginal dryness and dyspareunia can be treated with a topical estrogen cream, tablet, or vaginal ring, or with nonhormone moisturizing or lubrication products. In clinical trials, topical estrogen appears to be better than systemic estrogen for relieving these symptoms, and avoids high levels of circulating estrogen.

- In one clinical trial, an estradiol releasing silicone vaginal ring was also found to reduce the incidence of urinary tract infection.
- Systemic estrogen alone or with a progestin does not reduce incontinence, and in one large clinical trial Heart and Estrogen/Progestin Replacement Study (HERS) actually increased incontinence.

3. FRACTURES

3.1 Assessment

When discussing osteoporosis, it is important to be sure that the provider and the patient are using the same language. Some patients confuse osteoporosis (fragile bones) with osteoarthritis (painful joints). Other women (and some doctors) mistakenly believe that a diagnosis of osteoporosis means they should not exercise.

3.1.1 Risk Factors

Many factors are associated with an increased fracture risk in women, which may differ by fracture site. Most available data are on risk factors for spine or hip fractures in Caucasian women aged 65 and older. Predicting risk in younger women and other ethnic groups is less accurate.

Nonmodifiable risk factors for fractures:

- Age—there is an approximate doubling of fracture risk every 7 years
- Family history—history of osteoporotic fracture, especially hip fracture, in either parent or sibling approximately doubles the risk
- Personal history of osteoporotic fracture increases the risk twofold to fivefold
- Early menopause increases risk

Modifiable risk factors:

- Weight—there is an increased risk if thin, and a decreased risk if overweight
- Excessive weight loss is a powerful risk factor for bone loss and fracture

- Current smoking—increases the risk of all fractures
- Low calcium intake—increases the risk of hip fracture
- Vitamin D deficiency—can cause secondary hyperparathyroidism and osteoporosis
- Inadequate physical activity
- Factors associated with falls, some of which are modifiable:
 - Limited vision
 - Impaired cognition
 - Balance problems
 - Alcohol excess
 - Poor health, frailty, muscle weakness
 - Medications, particularly sedatives
 - Environmental hazards, such as poor lighting and loose area rugs
- Low bone density is a risk factor for fracture (fracture risk doubles for every 10–12 percent decrease in bone mineral density, a deviation of approximately—

 t-score or—1 z-score (measured by dual energy x ray absorptiometry [DEXA]).
- Many factors are associated with an increased fracture risk in women, which may differ by fracture site.

3.1.2 Case finding

Bone density testing is recommended in the United States for all women aged 65 years or older.

- DEXA of the hip is currently the gold standard for bone density measurements.
- Bone density results should be used in conjunction with information obtained in clinical risk assessment.

3.2 Prevention and Treatment

Better bones in old age are a function of peak bone mass (usually achieved around age 25), and subsequent rate of bone loss. Peak bone mass is maximized by an adequate calcium intake, physical activity, and not smoking. Lifestyle changes have been shown to improve bone density in young women and to prevent fractures in older women. They are low-cost, safe, and can be recommended widely.

3.2.1 Lifestyle

- Stop smoking
- Avoid extreme weight loss
- Add weight-bearing, muscle-building, and balance exercises
- · Avoid sedatives
- Avoid excess alcohol
- Correct visual impairment
- Fallproof the home

3.2.2 Diet

Lifestyle changes have been shown to improve bone density in young women and to prevent fractures in older women. • Correct calcium deficiency. A diet devoid of dairy products rarely provides more than 200 – 250 mg of calcium per day, which does not balance obligatory calcium loss and is associated with increased bone loss. Much of the bone loss can be attenuated by increasing calcium intake. Ideally, the combined diet and supplement intake should be 1,200 mg of calcium each day.

- In correcting dietary calcium deficiency, the first step is to increase calcium-rich foods; each dairy portion contains approximately 300 mg. Calcium-supplemented orange juice or mineral water rich in calcium are useful for women with lactose intolerance.
- If adequate dietary calcium is not likely, calcium supplements should be recommended. Calcium supplementation should be given concomitantly with Vitamin D. Clinical trials have shown that calcium with vitamin D can reduce fracture risk; no clinical trials have shown that vitamin D without calcium significantly reduces fracture risk. Clinical trials consistently show better bone preservation in women who take calcium with estrogen than estrogen alone.
- Supplement vitamin D intake for women 65 and older; 600 800 IU/day together with adequate calcium intake can reduce the risk of fracture in elderly women by about 25 percent.

3.2.3 Pharmacotherapy

With growing evidence for efficacy of osteoporosis treatments and with growing concern about drug costs, policymakers have recommended that expensive drugs not be used for osteoporosis prevention. In addition, all medications have risks and side effects. Therefore, aggressive pharmacotherapy should be reserved for women who are at high risk of fracture in the near future.

Practitioners in discussion with their patients must decide between therapy with bone-specific drugs or broad-spectrum drugs (HRT, selective estrogen receptor modulators).

• Drugs shown in clinical trials to prevent bone loss, that is, to be effective in prevention of osteoporosis, include estrogen, tibolone, raloxifene, alendronate, and risedronate. Drugs shown in clinical trials to prevent fractures include raloxifene, alendronate, and risedronate. These clinical trials were conducted in women at increased risk of fracture. Similar large trials have not been conducted using estrogen or tibolone. HERS found no difference in clinical fracture rate or height loss (a marker for vertebral fractures) in women assigned to HRT vs. placebo. Fracture was a preset secondary endpoint in HERS, but women in this trial were not at high risk for fracture.

Estrogen therapy

- Bone loss is accelerated during the first 5–10 years following menopause, and postmenopausal estrogen therapy is effective in preserving existing bone, whether begun in old age or at the time of the menopause. Results are similar with estrogen alone or when estrogen is used with a nonandrogenic progestin such as medroxyprogesterone acetate or progesterone. Androgenic progestins such as norethisterone-acetate have a synergistic activity when combined with 17-beta estradiol.
- Clinical trials have shown that doses of 0.3 mg per day of conjugated equine estrogen (lower than the previously recommended 0.625 mg per day), 0.5 mg of oral 17-beta estradiol or 25 micrograms of transdermal 17-beta estradiol maintain bone in most women when taken with adequate calcium. Smaller doses were better tolerated with regard to fewer episodes of uterine bleeding and less breast tenderness, two major reasons why women discontinue estrogen. Whether lower dosages of estrogen will be safer (lower rates of venous thomboembolism, breast cancer) remains to be proven.

When medication is indicated, based on a combination of clinical risk factors and low bone mineral density, the choice varies with the age of the patient and the severity of osteoporosis.

- Estrogen must be used continuously to preserve bone. Observational data suggest that bone density in older women who have never received estrogen is similar to bone density in women who used estrogen for 10 years and then discontinued it for another 10 years.
- Hormone treatment of women soon after menopause should be reserved for management of postmenopausal symptoms, but it will preserve bone in most women.

Non-estrogen therapy Asymptomatic women 10 or more years postmenopause, without severe osteoporosis, may prefer tibolone or raloxifene. For women aged 60 or older who have osteoporosis but are not at high risk for nonspine fracture, ralox-

Aggressive pharmacotherapy should be reserved for women who are at high risk of fracture in the near future.

ifene can be used to reduce the risk of spine fractures and for its possible other health benefits. Older women, particularly those with severe osteoporosis and prior fracture(s), may prefer alendronate or risedronate for their rapid acting bonespecific effects and reductions in nonspine as well as spine fractures.

Parathyroid hormone treatment by daily injection promises to be particularly effective for women with very severe osteoporosis who need to gain substantial amounts of bone.

The way medications are prescribed influences patient adherence. Beginning with a low dose for women prescribed estrogen helps reduce breast pain and slowly increasing dosage of raloxifene is useful in overcoming the hot flush side effect. Women who cannot tolerate the first medication often tolerate one of the other bone-sparing medications. A few trials have shown improved bone density over single therapy when a bisphosphonate is combined with estrogen or raloxifene, but there are no fracture data in women using these combinations (and the cost of combination therapy precludes routine use).

- Statins have been inconsistently associated with higher bone density. Statins have not been consistent in reducing bone loss or fracture risk, and these skeletal benefits have not been assessed in clinical trials.
- Clinical trial data show that a thiazide diuretic reduces but does not prevent bone loss.
- Observational studies suggest that people using thiazides are less likely to have fractures.

3.2.4 Complementary and Alternative Therapies

- Soy food (soy protein isolate) has been shown in clinical trials to have little or no benefit for the skeleton when ingested in the usually recommended amounts (20 – 25 grams of soy protein per day).
- Ipriflavone, a synthetic isoflavone, has been shown in preliminary studies to reduce bone loss but failed to improve bone density or reduce fracture risk in a large clinical study of women with osteoporosis.

4. CARDIOVASCULAR DISEASE

4.1 Assessment

When assessing a woman's knowledge about heart disease and stroke prevention, it is important to note that cardiovascular disease (CVD) is a more common cause of morbidity and mortality for women in most of the world than osteoporosis and cancer combined.

CVD is a more common cause of morbidity and mortality for women in most of the world than osteoporosis and cancer combined.

4.1.1 Risk Factors

The main risk factors for coronary heart disease (CHD) are: high blood cholesterol, high blood pressure, diabetes, and cigarette smoking. These same factors also apply to stroke and peripheral arterial disease, but the order of importance differs. High blood pressure is the most important risk factor for stroke, while

smoking has been consistently associated with peripheral arterial disease, and high blood cholesterol with CHD.

Nonmodifiable risk factors

- Age. For every 10-year increase in age, the risk for heart disease increases about threefold.
- The presence of CHD or other evidence of atherosclerotic arterial disease including stroke or lower extremity arterial disease increases the risk for myocardial infarction (MI) about fivefold. Atrial fibrillation, aortic stenosis, and narrowing of the coronary arteries are also risk factors for stroke.

- Family history of premature CHD (MI before age 55 in men, 65 in women) increases the risk for MI about twofold.
- The importance of a positive family history is amplified in women who smoke cigarettes.

Modifiable risk factors

- Cigarette smoking: Compared to smokers, nonsmokers or women who stop smoking have one-third the risk for MI.
- Physical activity: Women who walk briskly for 3 hours per week have a one-third lower risk for MI compared to women who do little exercise.
- Nutrition: Women whose usual diet is low in saturated and trans fats, and relatively high in unsaturated fats (including monounsaturated fats and fish oils), and high in cereal fiber, fruits, and vegetables, have half the risk for MI compared to women who do not have this healthy eating pattern.
- Weight: Lean women (body mass index below 25) have one-quarter less risk than overweight women, and less than one-half the risk of obese women (body mass index above 30).
- Fat distribution: Women with a waist circumference of less than 28 inches (71 cm) have one-third the risk for MI compared to women with a waist circumference of more than 38 inches (96.5 cm). Weight and weight distribution associated risks are not the same in all populations. For example, overweight and central obesity seem to be less important risk factors in African American women than in women of northern

European ancestry and Asian American women who seem to be at increased risk at lower weights.

- Psychosocial factors: Life stress situations, depression, and social isolation have been linked to increased risk for MI in women.
- Blood pressure: Women with a systolic blood pressure below 140 mmHg have a risk for MI one-half that of women with a level above 180 mmHg.
- Blood cholesterol: Women with a LDL cholesterol below 130 mg/dL (3.4 mmol/L) have a risk for MI one-half that of women with levels above 190 mg/dL (4.9 mmol/L).
- HDL cholesterol: Women with a HDL cholesterol level above 60 mg/dL (1.6 mmol/L) have a risk for MI which is one-third that of women with a level of less than 40 mg/dL (1.0 mmol/L) Contrary to popular opinion, women with high HDL cholesterol levels are not immune to MI.
- Triglycerides: Women with triglyceride levels below 150 mg/dL (1.7 mmol/L) have a risk of MI one-third lower than women with levels above 240 mg/dL (2.7 mmol/L).

The main risk factors for CHD are: high blood cholesterol, high blood pressure, diabetes, and cigarette smoking... • Diabetes: Women with diabetes by history or glycemia (fasting plasma glucose > 126 mg/dL (7.0 mmol/L) and/or 2 hour postchallenge glucose above 199 mg/dL (11.1 mmol/L) have a two- to fourfold increased risk of MI compared to women

without diabetes. About half of women with Type 2 diabetes do not know they have it. Diabetes is often first diagnosed when the patient has a MI.

4.1.2 Other Assessments

History

- Presence of any of the risk factors above.
- Symptoms compatible with transient ischemic attack, CHD, or lower extremity atherosclerosis.
- Use of HRT, antihypertensive drugs, lipid-lowering therapy, aspirin, and medication for diabetes.

Physical examination

- Pulses, auscultation for cardiac murmurs, and arterial bruits.
- Blood pressure at first visit. Women who have optimal blood pressure levels (<130/85 mmHg) are rechecked every 2 years (Europe: women > 40 years: every year), those with normal levels (<140/90 mmHg) are rechecked every year. Women with levels >140/90 mmHg need confirmation.
- Height, weight, waist circumference, calculate body mass index (BMI).

Laboratory tests

- Fasting glucose, total cholesterol, HDL cholesterol, triglycerides, calculated LDL cholesterol at first visit. For women without known CHD, the desirable lipid levels are LDL cholesterol <130 mg/dL (3.4 mmol/L), triglycerides <150 mg/dL (1.7 mmol/L), and HDL cholesterol levels >45 mg/dL (1.3 mmol/L).
- Repeat measurements that are normal every 5 years.
- Evidence does not support further screening for diabetes. Case finding may be appropriate in persons who have central obesity, high triglycerides, or a positive family history. Repeat glucose tests for women whose fasting blood glucose is elevated because the diagnosis of diabetes needs a confirmatory test. Women whose fasting plasma glucose is between 110 (6.1 mmol/L) and 126 mg/dL (7.0 mmol/L) or whose 2 hour glucose is between 140 (7.8 mmol/L) and 200 mg/dL (11.1 mmol/L) are at high risk of future diabetes.

• In some countries, homocysteine measurement is recommended.

Women with diabetes ... have a twoto fourfold increased risk of MI ... Other tests for CHD, cerebral arterial disease, and peripheral vascular disease as indicated by symptoms.

The combined effect of two or more risk factors is more powerful than any single risk factor, and some risk factors commonly occur together. (For example, screening for

diabetes may be most appropriate in persons with high blood pressure or high triglyceride levels.) When one risk factor is found, the presence of other factors should be sought and an assessment of overall risk should be made by counting the number of risk factors plus a 10-year risk assessment as in the National Cholesterol Education Program's Adult Treatment Panel III Report (NCEP ATP III) or as an assessment of the 10-year risk as recommended in European guidelines. Risk assessment can be used to motivate the patient to make lifestyle changes and comply with medication.

4.2 Prevention and Treatment

Even in the absence of clinical trial data, lifestyle and diet recommendations can be made to all women, not just those with heart disease risk factors or disease, because the diet, physical activity, and not smoking recommendations represent a return toward the evolutionary norm.

4.2.1 Lifestyle

• At each visit, reinforce nonsmoking status, or strongly encourage patient (and family) to stop smoking and avoid secondhand smoke. Prescribe counseling, nicotine replacement, or other pharmacotherapy as indicated in conjunction with behavioral therapy or a formal smoking cessation program.

- Encourage a minimum of 30 minutes of moderate-intensity dynamic exercise, e.g., brisk walking, at least 3 days a week, supplemented by an increase in daily lifestyle activities. Women who want to do more than the minimum should be encouraged to do so. Recommend medically supervised programs for women who have had a recent MI or revascularization procedure.
- Encourage gradual weight loss for overweight women through a combination of physical activity and portion control, healthy food choices, and recognition of triggers to overeating. Refer to weight loss support group or formal nutritional counseling when appropriate.
- Encourage positive coping mechanisms for stress (e.g., substitute physical activity for overeating or smoking in response to stressful life situations).

4.2.2 Diet

- Encourage a well-balanced and diversified eating pattern that is low in saturated fat and high in fresh fruits and vegetables and fiber. Prefer fats with higher monounsaturated content (e.g., olive oil, canola oil). Prefer seafood and skinless chicken to red meat. Prefer soft unsaturated margarine to hard margarine or butter. Use skim milk and skim milk products or at most 1 percent milk instead of products with a higher fat content. Limit the intake of high-cholesterol foods, avoid fast-food meals. Consume more than five servings of fruits and vegetables daily. Total dietary fiber intake from food should be 25–30 g per day.
- A clinical trial showed that eating fish two to three times per week reduced the risk of CVD.
- Encourage increased dietary consumption of omega-3 fatty acids.
- A clinical trial showed that a "Mediterranean diet," supplemented with alpha-linoleic acid, significantly reduced the risk of recurrent coronary events in patients with heart disease.

- Diets rich in antioxidant vitamins (i.e., nuts, fruits, and vegetables) are preferred over vitamin supplements.
- Limit salt intake to 6 g per day. A reduced salt/reduced saturated fat diet has been shown to reduce blood pressure in clinical trials.

Even in the absence of clinical trial data, lifestyle and diet recommendations can be made to all women ... • Prefer spices to salt in food preparation. Reduce intake of canned and commercial bakery goods, which are usually high in salt.

• Limit alcohol to less than one to two glasses per day: one glass equals 4 ounces of wine (approx. 120 mL),

12 ounces of beer (approx. 360 mL), or 1 1/2 ounces of 80-proof spirits (approx. 45 mL).

4.2.3 Pharmacotherapy

Blood pressure

 Achieve and maintain blood pressure <140/90 mmHg or lower if tolerated. If blood pressure remains above 140/90 mmHg after 3 months of reduced dietary salt, saturated fats and attempted weight loss, or if initial level is above 160/100 mmHg, initiate individualized pharmacotherapy. Goal blood pressure <130/80 mmHg if diabetic.

Beta-blockers, low-dose diuretics, and angiotensin converting enzyme (ACE) inhibitors have been shown in clinical trials to reduce the risk of MI in patients with high blood pressure.

Lipids and lipoproteins

- LDL cholesterol
 - (a) In women without CHD or CHD risk equivalents (other forms of atherosclerotic disease, diabetes, or 10-year risk more than 20 percent), the desirable LDL cholesterol level is <130 mg/dL (3.4 mmol/L).</p>

- If the LDL cholesterol level is >130 mg/dL (3.4 mmol/L) and two or more other risk factors are present, or the 10-year risk of MI is more than 10 percent, implement intensive lifestyle intervention and consider pharmacotherapy. If the 10-year risk is less than 10 percent, consider pharmacotherapy if the LDL cholesterol is >160 mg/dL (4.1 mmol/L).
- If the LDL cholesterol level is >190 mg/dL (4.9 mmol/L), pharmacotherapy is usually required.
- (b) In women with CHD or CHD equivalents, the desirable LDL cholesterol level is 100 mg/dL (2.6 mmol/L) or lower, and pharmacotherapy is generally required.
- Triglycerides and HDL cholesterol
 - If the triglycerides are >150 mg/dL (1.7 mmol/L) and the HDL cholesterol is below 40 mg/dL (1.0 mmol/L), treatment should still be aimed primarily at the LDL level. Lowering triglycerides and raising HDL levels become secondary targets of therapy and may influence the choice of drugs. Women with elevated triglycerides as their only lipid abnormality usually respond to intensive lifestyle measures. Some patients with very high triglyceride levels respond best to fibrates or niacin.
- Choice of drugs
 - Statins are the drugs of choice for high LDL cholesterol levels, irrespective of the levels of triglycerides or HDL cholesterol. Statins have been shown in clinical trials to reduce the risk of MI and stroke.

For women with moderate elevations of LDL cholesterol, raised triglycerides, and low HDL cholesterol levels, statins are the first choice. Fibrates and niacin have not been shown in randomized clinical trials (RCTs) to reduce CHD risk in women, although recommended in AHA/ACC (American Heart Association/ American College of Cardiology) guidelines for low HDL or high triglycerides. Start statin therapy promptly in patients with acute coronary syndrome.

HRT given orally reduces LDL cholesterol by 10 percent, raises HDL cholesterol by 10 percent, and raises triglycerides by 20 percent. HRT is not recommended for management of lipid disorders because of the lack of clinical trial evidence showing cardiovascular benefit.

Diabetes

- Target preprandial blood glucose in the range of 80-120 mg/dL (4.4-6.7 mmol/L), bedtime 100-140 mg/dL (5.5-7.8 mmol/L), Hgb A1c <7 percent
- Maintain LDL cholesterol <100 mg/dL (2.6 mmol/L) and triglycerides <150 mg/dL (1.7 mmol/L)
- Maintain blood pressure <130/80 mmHg (optimal <120/75 mmHg)
- In clinical trials, the initiation of oral HRT is accompanied by a two- to fourfold increased risk of venous thromboembolism and a small early increased risk of CHD and stroke. Based on observational studies, a reduction in risk for CHD after 2 or more years is possible, but the clinical trial evidence is lacking. Some experts see no reason to discontinue HRT in women who have been treated for many years, in view of the expected benefit for osteoporosis. Based on clinical trial data, an increased risk of venous thromboembolic disease persists for at least 4 years. The absolute increase in risk for venous thromboembolism is small—approximately two excess events in 8,000 treated women. This risk

may be reduced in women taking aspirin or statins.

- A history of venous thromboembolic disease is a contraindication to HRT.
- Low-dose aspirin can be recommended for women with established CVD (based on clinical trial data and probably for high-risk women [by inference only]). There is some concern that the risk benefit ratio may be different in women, who seem to have a higher risk of stroke than men. Consider clopidogrel or warfarin if aspirin contraindicated.
- Beta-blockers: If there are no contraindications (e.g., severe bradycardia, high degree heart block, acute heart failure, asthma, active peripheral vascular disease) start betablockers within hours of hospitalization for MI and acute coronary syndromes, or as soon as possible thereafter to lower the risk of reinfarction and of cardiac failure.
- ACE inhibitors. If there are no contraindications (e.g., renal artery stenosis, aortic stenosis, or severe hypotension), start ACE inhibitors within hours of hospitalization for MI, or as soon as possible thereafter, to lower the risk of reinfarction and of cardiac failure. Use ACE inhibitors to lower the risk of MI and death in patients with cardiac failure, left ventricular dysfunction, or high risk for CHD.

4.2.4 Complementary and Alternative Therapies

• Trials of vitamin E and beta-carotene supplements have failed to show benefit for CVD prevention.

5. CANCERS (BREAST, CERVIX, COLORECTAL, ENDOMETRIAL, OVARY, AND LUNG)

The major cancers that occur in postmenopausal women are breast, cervix, colorectal, endometrial, ovary, and lung.

In observational studies, the increased risk of breast cancer after 5 or more years of estrogen replacement therapy is similar to the risk associated with a delayed menopause or with obesity. In some of these studies, breast cancer risk was higher in women who used estrogen plus a progestin and higher in women who used estrogen plus progestin cyclically.

In observational studies, the increased risk of breast cancer after 5 or more years of estrogen replacement therapy is similar to the risk associated with a delayed menopause or with obesity. Endometrial cancer is associated with endogenous or unopposed exogenous estrogen levels. An increased risk of endometrial cancer occurs in menopausal women who have low levels of progestin to counterbalance the stimulating effect of estrogen on the endometrium. A 3-year clinical trial has shown that endometrial hyperplasia, a uterine cancer precursor, occurs in 10 percent of women for each year of unopposed estrogen use.

A number of observational epidemiological studies (including both prospective and case-control studies) have consistently shown that women on HRT have reduced risks of developing colorectal cancer or adenoma and of dying from colorectal cancer. There is weak evidence from observational studies that HRT increases the risk of ovarian and lung cancer.

Much of the increased risk for cancers in postmenopausal women can be linked to the effects of age and accumulated lifetime exposure to carcinogens.

5.1 Assessment

5.1.1 Risk Factors

Nonmodifiable risk factors

- Family history of breast, ovarian, or colorectal cancer, especially in a first degree relative.
- Age. Most cancer rates increase with age. The year-by-year increase in breast cancer rates persists but is less steep in women who do not take estrogen after the menopause.
- Previous history of cancer (invasive and in situ).
- Precursor lesions (benign proliferative breast disease, colorectal polyps, endometrial hyperplasia, and high grade squamous intra-epithelial lesions of the cervix infected with selected variants of human papillomavirus[HPV]).
- Reproductive and menstrual factors: Early menarche and late menopause increase risk for breast cancer, and possibly also endometrial and ovarian cancer. Early first pregnancy and multiparity decrease the risk for breast and ovarian cancers. Multiparity also reduces the risk of endometrial cancer.

Modifiable risk factors

- Estrogen treatment: Excess exogenous estrogen in the postmenopausal years increases risk for breast and endometrial cancers. It is unknown whether lower doses of estrogen will have different risks and benefits. Use of a progestin with the estrogen may increase breast cancer risk, but it decreases endometrial cancer risk if taken in an adequate regimen, either 10–14 days per month or daily. Past oral contraceptive use greater than 1 year decreases endometrial and ovarian cancer risk.
- Overweight: Heavier postmenopausal women are at increased risk for cancer of the breast, endometrium, and colon. The risk for breast cancer in obese women is comparable to that for long-term HRT.
- Nutrition: Women whose usual diet is low in fat and high in vegetables, fruits, and fiber have a reduced risk for colorectal and breast cancer. Recent clinical trials found no reduced risk of colon polyps, a cancer precursor, after either a high-fiber diet or a diet enriched with fruits and vegetables.
- Physical activity: Physically active women may have a reduced risk for colon cancer and possibly also breast and endometrial cancer.
- Cigarette smoking: Women smokers are at increased risk for lung, cervix, colorectal, oral, esophageal, and pancreatic cancers, as well as other less common epithelial cell cancers.
- Alcohol: Alcohol use increases the risk for breast (probably by increasing endogenous estrogen levels) and colon cancer, as well as more rare head and neck cancers.
- Radiation: High doses of radiation, used in the past for certain medical treatments, have been associated with increased risk for several cancers.

- Other exposures: Women infected with specific strains of HPV are at increased risk for cervical cancer.
- High radiographic density on mammogram carries about a twofold increased risk for breast cancer and may delay diagnosis by making mammograms harder to read. A clinical trial showed that this reversible condition occurs within 1 year in about 15 percent of most postmenopausal women treated with estrogen alone and in more than one-third of those treat-

Much of the increased risk for cancers in postmenopausal women can be linked to the effects of age and accumulated lifetime exposure to carcinogens.

ed with estrogen plus a progestin.

5.1.2 Case-finding

History

• Presence of any of the risk factors above

Physical examination

- Height, weight
- Clinical breast exam
- Pelvic exam
- Colorectal screening (can include fecal occult blood assay, flexible sigmoidoscopy, and colonoscopy).

Specific tests

The selection of screening modality and frequency will depend on individual and population prevalence of disease and available resources.^a

^a Horton R. Screening mammography—an overview revisited. Lancet 2001;358:1284–1285.

Olsen O, Gotzsche PC. Cochrane review on screening for breast cancer with mammography. Lancet 2001;358:1340–1342.

• Mammograms: There is disagreement about the benefit of mammograms before age 50, but women taking HRT and those with other risk factors may want to be tested. Mammograms are usually performed annually or biannually.

Despite much interest in complementary and alternative therapies, there are no trial data on their efficacy in reducing cancer risk. Because women's risk of breast cancer continues to increase with age, regular mammography screening remains appropriate even in old age although the benefit in women over age 75 has not been tested.

- Cervical smears: In countries where cervical cancer rates increase with age, cervical smears should be continued into old age.
- Pap smears have poor positive predictive value for postmenopausal women and do not have to be performed more often than every 2–3 years after a normal cytological result. The main advantage of an annual Pap smear is that it increases overall adherence to regular examination. Clinical observation shows no effect of hormone replacement therapy on cytologic abnormalities.
- Special methods that include testing for carcinogenic HPV strains reduce the number of falsepositive results and the attendant anxiety and cost. In the future, these tests may replace the Pap smear as the gold standard for early detection of cervical cancer.

5.2 Prevention and Treatment

Despite a great deal of information on factors that increase cancer risk, there are limited data on what can be done to reduce risk. There have been few clinical trials of prevention modalities. It is not clear that reversing a risk factor will reduce cancer risk. Nevertheless, some practical recommendations can be made.

5.2.1 Lifestyle

- Stop smoking.
- Limit alcohol use to less than one to two glasses per day: 1 glass equals 4 ounces of wine (approx. 120 mL), 12 ounces of beer (approx. 360 mL), or 1 1/2 ounces of 80-proof spirits (approx. 45 mL).
- Avoid unnecessary radiation.
- Avoid unopposed estrogen replacement therapy if uterus is present.
- Avoid HRT for more than 5 years except in presence of specific indications.
- Avoid postmenopausal weight gain.
- If overweight or obese, lose weight.
- Increase physical activity.

5.2.2 Diet

- Increase intake of vegetables, fruits, and fiber.
- Decrease fat and red meat intake.

5.2.3 Pharmacotherapy

Breast Cancer

- Consider tamoxifen use if high risk for breast cancer; a North American clinical trial found a reduced risk for breast cancer with 5-year use. Tamoxifen increases the risk of endometrial cancer, venous thromboembolism, and vasomotor symptoms.
- Although raloxifene is not approved for prevention or treatment of breast cancer, a 4-year trial of raloxifene in women with osteoporosis (not at high risk for breast cancer) showed a 90 percent risk reduction for estrogen-receptor positive breast cancer, and no increased risk of uterine cancer. Breast density is not increased with raloxifene use. A trial comparing raloxifene with tamoxifen is underway.

- When prescribing estrogen to women with an intact uterus, prescribe at least 10–14 days per month of progestin, to reduce endometrial cancer risk.
- Do not give progestin to women without a uterus; it is not necessary, and some observational studies suggest that estrogen plus progestin increases the risk of breast cancer more than estrogen alone.

5.2.4 Complementary and Alternative Therapies

• Despite much interest in complementary and alternative therapies, there are no trial data on their efficacy in reducing cancer risk. The most attractive candidate is soy protein, based on observational studies of low breast cancer risk in countries with high soy intake. • A trial of the effect of a low-fat, high-fiber diet on the recurrence of colorectal adenomas, a cancer precursor, failed to show the benefit of such a diet in reducing risk.

Other: Very-High-Risk Patients

- Consider removal of at-risk organs.
- Very high risk for breast cancer: additional screening by sonography.
- There is no trial evidence that additional screening such as pelvic ultrasound or Ca125 for women at high risk for ovarian cancer or beginning mammography before age 40 for women at high risk for breast cancer will improve the prognosis.
- Colonoscopy for women at high risk for colorectal cancer can be recommended based on clinical trial data.

6. DEMENTIA AND MENTAL HEALTH

For most disorders affecting the central nervous system, there are inadequate data upon which to base practice decisions. Alzheimer's disease merits particular mention, because it is common and a major concern of older women.

6.1 Assessment

6.1.1 Risk Factors for Alzheimer's Disease

• The only consistently identified risk factors are age, family history, and apolipoprotein E e4 allele.

For most disorders affecting the central nervous system, there are inadequate data upon which to base practice decisions.

- The risk of developing Alzheimer's disease doubles approximately every 5 years through the ninth decade of life.
- Uncommon forms of the illness that appear before the seventh decade of life are often transmitted as autosomal dominant disorders.
- Dominant inheritance is not characteristic of later-onset dementia, although family history remains a risk factor in this age group.
- Some observational studies suggest other risk factors for Alzheimer's disease, including prior history of head trauma, low educational achievement, presence of CHD, hypertension or hyperlipidemia, prior history of depression, and the absence of HRT. The evidence for these associations is inconsistent.

- Polypharmacy and thyroid disease are two examples of reversible causes of memory loss in older adults.
- Depressed mood is fairly common in persons with dementia; it may impair cognitive function or be a consequence of it.

6.1.2 Case-Finding for Suspected Dementia

- The medical, neurological, and psychiatric history should focus on potential causes of cognitive and behavioral change, including stroke, endocrine disease (e.g., thyroid disorders), toxic exposures (particularly the excessive use of psychotropic medications or medications with psychotropic side-effects), and depression.
- Family history should be assessed.
- Functional decline should be documented.
- The mental status examination should evaluate both cognition and mood. Commonly used tests for cognitive function are the Mini-Mental State Examination (MMSE) and the short Blessed Test for Orientation-Memory-Concentration. Standard validated questionnaires are available for testing for depressed mood in the elderly (e.g., the Beck Depression Inventory and the Geriatric Depression Scale).
- Laboratory assessment in the patient with dementia usually includes complete blood count; serum electrolytes and glucose; tests of renal, liver, and thyroid function; and B-12 level. Screening for syphilis and HIV should be considered in at-risk populations.

- Brain imaging study (CT scan or MRI scan) is often used to exclude space-occupying lesions, evaluate suspected cerebrovascular disease, or evaluate suspected hydrocephalus. The diagnostic yield for this procedure is low if the neurological examination is normal and the history and examination are otherwise typical for Alzheimer's disease.
- A common cause of cognitive impairment in the elderly is overmedication. A careful review of all medications taken by the patient may lead to the identification of a reversible cause of confusion or memory loss.

6.2 Prevention and treatment

• There are no proven preventive measures for Alzheimer's disease. For prevention of vascular dementia, it is reasonable to follow preventive recommendations listed for CVD.

6.2.1 Lifestyle

- Ensure a safe, stable, and structured environment.
- Encourage social interventions such as power of attorney and caregiver respite as appropriate.

6.2.2 Diet

- Diet should be well-balanced.
- Discourage excess alcohol use.
- Discourage smoking, which can pose a fire hazard.

6.2.3 Pharmacotherapy

- Reduce unnecessary or optional medications.
- Identify and treat depression and other behavioral disturbances when they are distressing to patients or hinder their care.
- There is clinical trial evidence that drugs that inhibit the breakdown of acetylcholine in the brain are often of mild symptomatic benefit.
- There is no clinical trial evidence that estrogen improves symptoms or delays symptomatic progression in women with Alzheimer's disease.
- There are no published long-term clinical trial data on potential effects of HRT on age-associated cognitive decline.
- One clinical trial found that vitamin E slows progression without improving cognition in patients with moderate dementia due to Alzheimer's Disease.
- One clinical trial of antihypertensive medication in cognitively intact older adults with hypertension showed a small but significant difference in the rate of memory loss and the incidence of dementia in those on active treatment.
- One clinical trial found raloxifene was associated with less cognitive decline in two cognitive function tests in older women.

6.2.4 Complementary and Alternative Therapies

• There is limited trial evidence that gingko biloba may offer mild cognitive benefit when given to patients with dementia.

7. CONCLUSIONS

In the last 20 years, menopause has become a household word, with much better understanding of its consequences. The growing numbers of postmenopausal women and clinical trials have coincided to draw increasing attention to the perimenopausal and postmenopausal years. Better studies of older therapies and the expanded number of new choices today, with more in development and evaluation, have complicated provider and patient choices, but greatly improved the potential for effective intervention.

SUGGESTED READING (SYSTEMATIC REVIEWS AND RECENT CLINICAL TRIALS ONLY)

Menopause Transition

Albertazzi P, Pansini F, Bonaccorsi G, Zanotti L, Forini E, De Aloysio D. The effect of dietary soy supplementation on hot flushes. Obstet Gynecol 1998;1:6–11.

Anderson RU, Mobley D, Blank B, Saltzstein D, Susset J, Brown JS. Once daily controlled versus immediate release oxybutynin chloride for urge urinary incontinence. OROS Oxybutynin Study Group. J Urol 1999;161:1809–1812.

Brzezinski A, Adlercreutz H, Shaoul R. et al. Short-term effects of phytoestrogen-rich diet on postmenopausal women. Menopause 1997;4:89–94.

Burgio KL, Locher JL, Goode PS, et al. Behavioral vs. drug treatment for urge urinary incontinence in older women: a randomized controlled trial. JAMA 1998;280:1995–2000.

Chenoy R, Hussain S, Tayob Y, O'Brien PM, Moss MY, Morse PF. Effect of oral gamolenic acid from evening primrose oil on menopausal flushing. Br Med J 1994;308:501–503.

Coope J, Williams S, Patterson JS. A study of the effectiveness of propranolol in menopausal hot flushes. Br J Obstet Gynaecol 1978;85:472–475.

Eriksen B. A randomized, open, parallel-group study on the preventive effect of an estradiol-releasing vaginal ring (Estring) on recurrent urinary tract infections in postmenopausal women. Am J Obstet Gynecol 1999;180:1072–1079.

Grady D, Brown JS, Vittinghoff E, Applegate W, Varner E, Snyder T. Postmenopausal hormones and incontinence: The Heart and Estrogen/Progestin Replacement Study. Obstet Gynecol 2001;97:116–120.

Greendale GA, Reboussin BA, Hogan P, et al. Symptom relief and side effects of postmenopausal hormones: results from the Postmenopausal Estrogen/Progestin Interventions Trial. Obstet Gynecol 1998;92:982–988. Hirata JD, Swiersz LM, Zell B, Small R, Ettinger B. Does dong quai have estrogenic effects in postmenopausal women? A double-blind, placebo-controlled trial. Fertil Steril 1997;68:981–986.

Lopes P, Merkus HM, Nauman J, Bruschi F, Foidart JM, Calaf J. Randomized comparison of intranasal and transdermal estradiol. Obstet Gynecol 2000;96:906–912.

Loprinzi CL, Kugler JW, Sloan JA, et al. Venlafaxine in management of hot flashes in survivors of breast cancer: a randomised controlled trial. Lancet 2000;356:2059-2063.

Loprinzi CL, Michalak JC, Quella SK, et al. Megestrol acetate for the prevention of hot flashes. N Engl J Med 1994;331:347–352.

Pandya KJ, Raubertas RF, Flynn PJ, et al. Oral clonidine in postmenopausal patients with breast cancer experiencing tamoxifen-induced hot flashes: a University of Rochester Cancer Center Community Clinical Oncology Program study. Ann Intern Med 2000;132:788–793.

Raz R, Stamm WE. A controlled trial of intravaginal estriol in postmenopausal women with recurrent urinary tract infections. N Engl J Med 1993;329:753–756.

Rebar RW, Trabal J, Mortola J. Low-dose esterified estrogens (0.3 mg/day): long-term and short-term effects on menopausal symptoms and quality of life in postmenopausal women. Climacteric 2000;3:176–182.

Rymer J, Morris EP. Extracts from "Clinical Evidence": Menopausal symptoms. BMJ 2000;321:1516–1519.

Saletu B, Brandstatter N, Metka M, et al. Double-blind, placebo-controlled, hormonal, syndromal and EEG mapping studies with transdermal oestradiol therapy in menopausal depression. Psychopharmacology 1995;122:321–329. Sirtori CR, Pazzucconi F, Colombo L, Battistin P, Bondioli A, Descheemaeker K. Double-blind study of the addition of high-protein soya milk v. cows' milk to the diet of patients with severe hypercholesterolaemia and resistance to or intolerance of statins. Br J Nutr 1999;82:91–96.

St. Germain A, Peterson CT, Robinson JG, Alekel DL. Isoflavone-rich or isoflavone-poor soy protein does not reduce menopausal symptoms during 24 weeks of treatment. Menopause 2001;8:17–26.

Stearns V, Isaacs C, Rowland J, et al. A pilot trial assessing the efficacy of paroxetine hydrochloride (Paxil) in controlling hot flashes in breast cancer survivors. Ann Oncol 2000;11:17–22.

Utian WH, Shoupe D, Bachmann G, Pinkerton JV, Pickar JH. Relief of vasomotor symptoms and vaginal atrophy with lower doses of conjugated equine estrogens and medroxyprogesterone acetate. Fertil Steril 2001;75:1065–1079.

Fractures

Alexandersen P, Toussaint A, Christiansen C, et al. Ipriflavone in the treatment of postmenopausal osteoporosis: a randomized controlled trial. JAMA 2001;285:1482–1488.

Bjarnason NH, Bjarnason K, Haarbo J, Rosenquist C, Christiansen C. Tibolone: prevention of bone loss in late postmenopausal women. J Clin Endocrinol Metab 1996;81:2419–2422.

Black DM, Cummings SR, Karpf DB, et al. Randomised trial of effect of alendronate on risk of fracture in women with existing vertebral fractures. Fracture Intervention Trial Research Group. Lancet 1996;348:1535–1541.

Black DM, Thompson DE, Bauer DC, et al. Fracture risk reduction with alendronate in women with osteoporosis: the Fracture Intervention Trial (FIT) Research Group. J Clin Endocrinol Metab 2000;85:4118–4124.

Cauley JA, Black DM, Barrett-Connor E, et al. Effects of hormone replacement therapy on clinical fracture and height loss: The Heart and Estrogen/Progestin Replacement Study (HERS). Am J Med 2001;110:442–450.

Chapuy MC, Arlot ME, Duboeuf F, et al. Vitamin D3 and calcium to prevent hip fractures in the elderly women. N Engl J Med 1992;327:1637–1642.

Cummings SR, Black D, Thompson DE, et al. Effect of alendronate on risk of fracture in women with low bone density but without vertebral fractures—results from the Fracture Intervention Trial. JAMA 1998:280:2077–2082.

Effects of hormone therapy on bone mineral density: results from the Postmenopausal Estrogen/Progestin Interventions (PEPI) trial. The Writing Group for the PEPI Trial. JAMA 1996;276:1389–1396.

Elders PJ, Netelenbos JC, Lips P, et al. Calcium supplementation reduces vertebral bone loss in perimenopausal women: a controlled trial in 248 women between 46 and 55 years of age. J Clin Endocrinol Metab 1991;73:533–540.

Ettinger B, Black DM, Mitlak BH, et al. Reduction of vertebral fracture risk in postmenopausal women with osteoporosis treated with raloxifene: results from a 3-year randomized clinical trial. Multiple Outcomes of Raloxifene Evaluation (MORE) Investigators. JAMA 1999;282:637–645.

Field-Munves E. Evidence-based decisions for the treatment of osteoporosis. Ann Long-term Care 2001;9:70–79.

Gillespie WJ, Henry DA, O'Connell DL, Robertson J. Vitamin D and Vitamin D analogues for preventing fractures associated with involutional and post-menopausal osteoporosis. Cochrane Library 1999;1:1–17.

Harris ST, Watts NB, Genant HK, et al. Effects of risedronate treatment on vertebral and nonvertebral fractures in women with postmenopausal osteoporosis: a randomized controlled trial. Vertebral Efficacy with Risedronate Therapy (VERT) Study Group. JAMA 1999;282:1344–1352.

Heikkinen J, Vaheri R, Kainulainen P, Timonen U. Long-term continuous combined hormone replacement therapy in the prevention of postmenopausal bone loss: a comparison of high- and low-dose estrogen-progestin regimens. Osteoporos Int 2000;11:929–937.

Hosking D, Chilvers CE, Christiansen C, et al. Prevention of bone loss with alendronate in postmenopausal women under 60 years of age. Early Postmenopausal Intervention Cohort Study Group. N Engl J Med 1998;338:485–492.

Komulainen MH, Kroger H, Tuppurainen MT, et al. HRT and Vit D in prevention of non-vertebral fractures in postmenopausal women; a 5 year randomized trial. Maturitas 1998;31:45–54.

LaCroix AZ, Ott SM, Ichikawa L, Scholes D, Barlow

WE. Low-dose hydrochlorothiazide and preservation of bone mineral density in older adults. A randomized, double-blind, placebo-controlled trial. Ann Intern Med 2000;133:516–526.

Liberman UA, Weiss SR, Broll J, et al. Effect of alendronate on bone mineral density and the incidence of fractures in postmenopausal osteoporosis. The Alendronate Phase III Osteoporosis Treatment Study Group. N Engl J Med 1995;333:1437–1443.

Lindsay R, Cosman F, Lobo RA, et al. Addition of alendronate to ongoing hormone replacement therapy in the treatment of osteoporosis: a randomized, controlled clinical trial. J Clin Endocrinol Metab 1999;84:3076–3081.

Marcus R. Role of exercise in preventing and treating osteoporosis [review]. Rheum Dis Clin North Am 2001;27:131-141.

McClung M, Clemmesen B, Daifotis A, et al. Alendronate prevents postmenopausal bone loss in women without osteoporosis. A double-blind, randomized, controlled trial. Alendronate Osteoporosis Prevention Study Group. Ann Intern Med 1998;128:253–261.

Nieves JW, Komar L, Cosman F, Lindsay R. Calcium potentiates the effect of estrogen and calcitonin on bone mass: review and analysis [review]. Am J Clin Nutr 1998;67:18–24.

Pols HA, Felsenberg D, Hanley DA, et al. Multinational, placebo-controlled, randomized trial of the effects of alendronate on bone density and fracture risk in postmenopausal women with low bone mass: results of the FOSIT study. Foxamax International Trial Study Group. Osteoporos Int 1999;9:461–468.

Potter SM, Baum JA, Teng H, Stillman RJ, Shaw NF, Erdman JW, Jr. Soy protein and isoflavones: their effects on blood lipids and bone density in postmenopausal women. Am J Clin Nutr 1998;68(Suppl 6):1375S–1379S.

Recker RR, Davies KM, Dowd RM, Heaney RP. The effect of low-dose continuous estrogen and progesterone with calcium and vitamin D on bone in elderly women. A randomized, controlled trial. Ann Intern Med 1999;130:897–904.

Reid IR. Pharmacological management of osteoporosis in postmenopausal women: a comparative review. Drugs Aging 1999;15:349–363. Ringsberg KA, Gardsell P, Johnell O, Josefsson PO, Obrant KJ. The impact of long-term moderate physical activity on functional performance, bone mineral density and fracture incidence in elderly women. Gerontology. 2001;47:15–20.

Schneider DL, Barrett-Connor EL, Morton DJ. Timing of postmenopausal estrogen for optimal bone mineral density—the Rancho-Bernardo Study. JAMA 1997;277:534–547.

Wangen KE, Duncan AM, Merz-Demlow BE, et al. Effects of soy isoflavones on markers of bone turnover in premenopausal and postmenopausal women. J Clin Endocrinol Metab 2000;85:3043–3048.

Cardiovascular Disease

Angerer P, Störk S, Kothny W, Schmitt P, von Schacky C. Effect of oral postmenopausal hormone replacement on progression of atherosclerosis. a randomized, controlled trial. Arterioscler Thromb Vasc Biol 2001;21:262–268.

Appel LJ, Espeland MA, Easter L, Wilson AC, Folmar S, Lacy CR. Effects of reduced sodium intake on hypertension control in older individuals: results from the Trial of Nonpharmacologic Interventions in the Elderly (TONE). Arch Intern Med 2001;161:685–693.

Burr ML, Fehily AM, Gilbert JF, et al. Effect of changes in fat, fish, and fibre intakes on death and myocardial infarction: diet and reinfarction trial (DART). Lancet 1989;2:757–761.

Burr ML, Holliday RM, Fehily AM, Whitehead PJ. Haematological prognostic indices after myocardial infarction: evidence from the diet and reinfarction trial (DART). Eur Heart J 1992;13:166-170.

Executive Summary of the Third Report of the National Cholesterol Education Program (NCEP) Expert Panel on Detection, Evaluation, and Treatment of High Blood Cholesterol in Adults (Adult Treatment Panel III). JAMA 2001;285:2486–2497.

Faergeman O. Hypertriglyceridemia and the fibrate trials [review]. Curr Opin Lipidol 2000;11:609–614.

Grady D, Wenger NK, Herrington D, et al. Postmenopausal hormone therapy increases risk for venous thromboembolic disease—the Heart and Estrogen/progestin Replacement Study. Ann Intern Med 2000;132:689–696. Hansson L, Zanchetti A, Carruthers SG, et al. Effects of intensive blood pressure lowering and low-dose aspirin in patients with hypertension: principal results of the Hypertension Optimal Treatment (HOT) randomised trial. HOT Study Group. Lancet 1998;351:1755–1762.

Hebert PR, Gaziano JM, Chan KS, Hennekens CH. Cholesterol lowering with statin drugs, risk of stroke, and total mortality. An overview of randomized trials. JAMA 1997;278:313–321.

Hemminki R, McPherson K. Value of drug-licensing documents in studying the effect of postmenopausal hormone therapy on cardiovascular disease. Lancet 2000;355:566–569.

Herrington DM, Reboussin DM, Brosnihan KB, et al. Effects of estrogen replacement on the progression of coronary artery atherosclerosis. N Engl J Med 2000:343:522–529.

Hulley S, Grady D, Bush T, et al. Randomized trial of estrogen plus progestin for secondary prevention of coronary heart disease in postmenopausal women. Heart and Estrogen/progestin Replacement Study (HERS) Research Group. JAMA 1998;280:605-612.

Kuller LH, Simkin-Silverman LR, Wing RR, Meilahn EN, Ives DG. Women's Healthy Lifestyle Project: a randomized clinical trial.—Results at 54 months. Circulation 2001;103:32–37.

Lancaster T, Stead L, Silagy C, Sowden A. Effectiveness of interventions to help people stop smoking: findings from the Cochrane Library. BMJ 2000;321:355–358.

LaRosa JC, He J, Vupputuri S. Effect of statins on risk of coronary disease: a meta-analysis of randomized controlled trials. JAMA 1999;282:2340–2346.

Lenfant C. Statement on preliminary trends in the Women's Health Initiative. NHLBI Communications Office. Available at:

http://www.nhlbi.nih.gov/whi/index.html. Accessed at 25 October 2001.

Lieberman EH, Gerhard MD, Uehata A, et al. Estrogen improves endothelium-dependent, flow-mediated vasodilation in postmenopausal women. Ann Intern Med 1994;121:936–941. Liu L, Wang JG, Gong L, Liu G, Staessen JA. Comparison of active treatment and placebo in older Chinese patients with isolated systolic hypertension. Systolic Hypertension in China (Syst-China) Collaborative Group. J Hypertens 1998;16(12 Pt 1):1823–1829.

Low-dose aspirin and vitamin E in people at cardiovascular risk: a randomised trial in general practice— Collaborative Group of the Primary Prevention Project. Lancet 2001;357:89–95.

Packer M, Bristow MR, Cohn JN, et al. The effect of carvedilol on morbidity and mortality in patients with chronic heart failure. U.S. Carvedilol Heart Failure Study Group. N Engl J Med 1996;334:1349–1355.

Robertson RM, Smaha L. Can a Mediterranean-style diet reduce heart disease? Circulation. 2001;103:1821–1822.

Schwartz GG, Olsson AG, Ezekowitz MD, et al. Effects of atorvastatin on early recurrent ischemic events in acute coronary syndromes: the MIRACL study: a ran-domized controlled study. JAMA 2001;285:1711–1718.

Simon JA, Hsia J, Cauley JA, et al. Postmenopausal hormone therapy and risk of stroke: the Heart and Estrogen-progestin Replacement Study (HERS). Circulation 2001;103:638–642.

SHEP Cooperative Research Group. Prevention of stroke by antihypertensive drug treatment in older persons with isolated systolic hypertension: final results of the Systolic Hypertension in the Elderly Program (SHEP). JAMA 1991;265:3255–3264.

Staessen JA, Fagard R, Thijs L, et al. Randomised double-blind comparison of placebo and active treatment for older patients with isolated systolic hypertension. The Systolic Hypertension in Europe (Syst-Eur) Trial Investigators. Lancet 1997;350:757–764.

The Acute Infarction Ramipril Efficacy (AIRE) Study Investigators. Effect of ramipril on mortality and morbidity of survivors of acute myocardial infarction with clinical evidence of heart failure—the Acute Infarction Ramipril Efficacy (AIRE) Study Investigators. Lancet 1993;342:821-828.

The Trials of Hypertension Prevention Collaborative Research Group. Effects of weight loss and sodium reduction intervention on blood pressure and hypertension incidence in overweight people with high-normal blood pressure: the Trials of Hypertension Prevention, phase 2. Arch Intern Med 1997;157:657–667. Virtamo J, Rapola JM, Ripatti S, et al. Effect of vitamin E and beta carotene on the incidence of primary nonfatal myocardial infarction and fatal coronary heart disease. Arch Intern Med 1998;158:668–675.

Viscoli CM, Brass LM, Kernan WN, Sarrel PM, Suissa S, Horwitz RI. A clinical trial of estrogen-replacement therapy after ischemic stroke. N Engl J Med 2001;345:1243-1249.

Whelton PK, Appel LJ, Espeland MA, et al. Sodium reduction and weight loss in the treatment of hypertension in older persons: a randomized controlled trial of nonpharmacologic interventions in the elderly (TONE). Tone Collaborative Research Group. JAMA 1998;279:839–846.

Yusuf S, Sleight P, Pogue J, Bosch J, Davies R, Dagenais G. Effects of an angiotensin-convertingenzyme inhibitor, ramipril, on cardiovascular events in high-risk patients. The Heart Outcomes Prevention Evaluation Study Investigators [published erratum appears in N Engl J Med 2000;342:748 and 1376]. N Engl J Med 2000;342:145–153.

Cancers

Archer DF, Dorin M, Lewis V, Schneider DL, PickarJH. Effects of lower doses of conjugated equine estrogens and medroxyprogesterone acetate on endometrial bleeding. Fertil Steril 2001;75:1080–1087.

Boyd NF, Byng JW, Jong RA, et al. Quantitative classification of mammographic densities and breast cancer risk: results from the Canadian National Breast Screening Study. J Natl Cancer Inst 1995;87:670–675.

Bush TL, Whiteman M, Flaws JA. Hormone replacement therapy and breast cancer: a qualitative review. Obstet Gynecol 2001;98:498–508.

Collaborative Group on Hormonal Factors in Breast Cancer. Breast cancer and hormone replacement therapy: collaborative reanalysis of data from 51 epidemiological studies of 52,705 women with breast cancer and 108,411 women without breast cancer. Lancet 1997;350:1047–1059.

Cummings SR, Eckert S, Krueger KA, et al. The effect of raloxifene on risk of breast cancer in postmenopausal women: results from the MORE randomized trial. Multiple Outcomes of Raloxifene Evaluation [published erratum appears in JAMA 1999;282:2124]. JAMA 1999 281:2189-2197. Fisher B, Dignam J, Bryant J, Wolmark N. Five versus more than five years of tamoxifen for lymph node-negative breast cancer: updated findings from the National Surgical Adjuvant Breast and Bowel Project B-14 randomized trial. J Natl Cancer Inst 2001;93:684–690.

Greendale GA, Reboussin BA, Sie A, et al. Effects of estrogen and estrogen-progestin on mammographic parenchymal density. Postmenopausal Estrogen/ Progestin Interventions (PEPI) Investigators. Ann Intern Med 1999;130:262–269.

Lieberman DA, Weiss DG. One-time screening for colorectal cancer with combined fecal occult-blood testing and examination of the distal colon. N Engl J Med 2001;345:555–560.

Lobo RA, Bush T, Carr BR, Pickar JH. Effects of lower doses of conjugated equine estrogens and medroxyprogesterone acetate on plasma lipids and lipoproteins, coagulation factors, and carbohydrate metabolism. Fertil Steril 2001;76:13–24.

Lobo RA and Whitehead MI. Is low-dose hormone replacement therapy for postmenopausal women efficacious and desirable? Climacteric 2001;4:110–119.

Manos MM, Kinney WK, Hurley LB, et al. Identifying women with cervical neoplasia: using human papillomavirus DNA testing for equivocal Papanicolaou results. JAMA 1999;281:1605–1610.

Pickar JH, Yeh IT, Wheeler JE, Cunnane MF, Speroff L. Endometrial effects of conjugated equine estrogens and medroxyprogesterone acetate. Fertil Steril 2001;76:25–31.

Sawaya GF, Grady D, Kerlikowske K, et al. The positive predictive value of cervical smears in previously screened postmenopausal women: the Heart and Estrogen/progestin Replacement Study (HERS). Ann Intern Med 2000;133:942–950.

Schatzkin A, Lanza E, Corle D, et al. Lack of effect of a low-fat, high-fiber diet on the recurrence of colorectal adenomas. Polyp Prevention Trial Study Group. N Engl J Med 2000;342:1149–1155.

Utian WH, Shoupe D, Bachmann G, Pinkerton JV, Pickar JH. Relief of vasomotor symptoms and vaginal atrophy with lower doses of conjugated equine estrogens and medroxyprogesterone acetate. Fertil Steril 2001;76:1065–1079.

Dementia and Mental Health

Forette F, Seux ML, Staessen JA, et al. Prevention of dementia in randomised double-blind placebocontrolled Systolic Hypertension in Europe (Syst-Eur) trial. Lancet 1998;352:1347–1351.

Henderson VW, Paganini-Hill A, Miller BL, et al. Estrogen for Alzheimer's disease in women: randomized, double-blind, placebo-controlled trial. Neurology 2000;54:295–301.

Le Bars PL, Katz MM, Berman N, Itil TM, Freedman AM, Schatzberg AF. A placebo-controlled, doubleblind, randomized trial of an extract of ginkgo biloba for dementia. JAMA 1997;278:1327–1332.

LeBlanc ES, Janowsky J, Chan BK, Nelson HD. Hormone replacement therapy and cognition: systematic review and meta-analysis. JAMA 2001;285:1489–1499.

Mulnard RA, Cotman CW, Kawas C, et al. Estrogen replacement therapy for treatment of mild to moderate Alzheimer Disease: a randomized controlled trial. JAMA 2000;283:1007–1015. Raskind MA, Peskind ER, Wessel T, Yuan W. Galantamine in AD: a 6-month randomized, placebocontrolled trial with a 6-month extension. Neurology 2000;54:2261–2268.

Rogers SL, Farlow MR, Doody RS, et al. A 24-week, double-blind, placebo-controlled trial of donepezil in patients with Alzheimer's disease. Neurology 1998;50:136–145.

Rösler M, Anand R, Cicin-Sain A, et al. Efficacy and safety of rivastigmine in patients with Alzheimer's disease: international randomised controlled trial. Br Med J 1999;318:633–640.

Sano M, Ernesto C, Thomas RG, et al. A controlled trial of selegiline, alpha-tocopherol, or both as treatment for Alzheimer's disease. The Alzheimer's Disease Cooperative Study. New England Journal of Medicine 1997;336:1216–1222.

Wang PN, Liao SQ, Liu RS, et al. Effects of estrogen on cognition, mood, and cerebral blood flow in AD: a controlled study. Neurology 2000;54:2061–2066.

Yaffe K, Krueger K, Sarkar S, et al. Cognitive function in postmenopausal women treated with raloxifene. N Engl J Med 2001;344:1207–1213.

Discrimination Prohibited: Under provisions of applicable public laws enacted by Congress since 1964, no person in the United States shall, on the grounds of race, color, national origin, handicap, or age, be excluded from participation in, be denied the benefits of, or be subjected to discrimination under any program or activity (or, on the basis of sex, with respect to any education program or activity) receiving Federal financial assistance. In addition, Executive Order 11141 prohibits discrimination on the basis of age by contractors and subcontractors in the performance of Federal contracts, and Executive Order 11246 states that no federally funded contractor may discriminate against any employee or applicant for employment because of race, color, religion, sex, or national origin. Therefore, the National Heart, Lung, and Blood Institute must be operated in compliance with these laws and Executive Orders.

U.S. DEPARTMENT OF HEALTH AND HUMAN SERVICES

Public Health Service National Institutes of Health National Heart, Lung, and Blood Institute

For Administrative Use Only March 2002